

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL088823

Key Points:

- A widespread pulse of aged dissolved organic carbon (DOC) occurred in the Mackenzie River and its tributaries in June 2018
- Export of aged DOC is consistent with a prolonged warming period and the formation of supra-permafrost taliks
- Mobilization of aged DOC and nitrate suggests percolation of supra-permafrost groundwater through previously frozen soil layers

Supporting Information:

- Supporting Information S1
- Table S1

Correspondence to:

M. S. Schwab, melissa.schwab@erdw.ethz.ch

Citation:

Schwab, M. S., Hilton, R. G., Raymond, P. A., Haghipour, N., Amos, E., Tank, S. E., et al. (2020). An abrupt aging of dissolved organic carbon in large Arctic rivers. *Geophysical Research Letters*, *47*, e2020GL088823. https://doi. org/10.1029/2020GL088823

Received 18 MAY 2020 Accepted 14 OCT 2020 Accepted article online 4 NOV 2020

Author Contributions:

Conceptualization: Melissa S. Schwab, Robert G. Hilton, Timothy I. Eglinton Data curation: Melissa S. Schwab, Robert M. Holmes Formal analysis: Melissa S. Schwab Investigation: Melissa S. Schwab, Robert G. Hilton, Edwin Amos, Negar Haghipour Supervision: Robert G. Hilton, Timothy I. Eglinton Visualization: Melissa S. Schwab Writing - original draft preparation: Melissa S. Schwab, Robert G. Hilton (continued)

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

An Abrupt Aging of Dissolved Organic Carbon in Large Arctic Rivers

Melissa S. Schwab¹, Robert G. Hilton², Peter A. Raymond³, Negar Haghipour^{1,4}, Edwin Amos⁵, Suzanne E. Tank⁶, Robert M. Holmes⁷, Edward T. Tipper⁸, and Timothy I. Eglinton¹

¹Department of Earth Sciences, ETH Zurich, Zurich, Switzerland, ²Department of Geography, Durham University, Durham, UK, ³Yale School of Forestry and Environmental Studies, Yale University, New Haven, CT, USA, ⁴Laboratory of Ion Beam Physics, ETH Zurich, Zurich, Switzerland, ⁵Aurora Research Institute, Inuvik, Northwest Territories, Canada, ⁶Department of Biological Sciences, University of Alberta, Edmonton, Alberta, Canada, ⁷Woods Hole Research Center, Falmouth, MA, USA, ⁸Department of Earth Sciences, University of Cambridge, Cambridge, UK

Abstract Permafrost thaw in Arctic watersheds threatens to mobilize hitherto sequestered carbon. We examine the radiocarbon activity ($F^{14}C$) of dissolved organic carbon (DOC) in the northern Mackenzie River basin. From 2003–2017, DOC- $F^{14}C$ signatures (1.00 ± 0.04 ; n = 39) tracked atmospheric ${}^{14}CO_2$, indicating export of "modern" carbon. This trend was interrupted in June 2018 by the widespread release of aged DOC (0.85 ± 0.16 , n = 28) measured across three separate catchment areas. Increased nitrate concentrations in June 2018 lead us to attribute this pulse of ${}^{14}C$ -depleted DOC to mobilization of previously frozen soil organic matter. We propose export through lateral perennial thaw zones that occurred at the base of the active layer weakened by preceding warm summer and winter seasons. Although we are not yet able to ascertain the broader significance of this "anomalous" mobilization event, it highlights the potential for rapid and large-scale release of aged carbon from permafrost.

Plain Language Summary The thaw of continuously frozen grounds in the Arctic induced by regional warming accelerates the release of carbon to the atmosphere and river systems. Of particular concern is the fate of dissolved organic carbon (DOC) due to its potential for rapid oxidation to carbon dioxide. In order to understand the ramifications of a warming climate, we analyze the radiocarbon age of DOC in the northern Mackenzie River—a major Arctic river basin. DOC in large Arctic rivers has been characterized by young radiocarbon ages, from modern vegetation and surface soils. In June 2018, we recorded a departure from long-term observations: Older DOC was measured in three large catchments draining into the Mackenzie Delta. This release of aged DOC followed a warm summer and the second warmest winter on record. We infer that the aged DOC derived from thaw of deeper soil horizons and subsequent carbon mobilization and riverine export. This is the first time such an event has been documented; it highlights the potential for abrupt and widespread aged DOC export with important implications for regional and global carbon cycles.

1. Introduction

The northern circumpolar permafrost regions contain 44% of the global soil organic carbon stock within its top 3 m, corresponding to approximately twice the amount of carbon present in the atmosphere (Hugelius et al., 2014; Schuur et al., 2008). Organic-rich permafrost soils have accumulated over millennia in peat deposits and deltaic sediments (Zimov et al., 2006) with subzero temperatures acting to inhibit microbial decomposition and preserve organic matter (Ping et al., 2011). Warming can lead to thawing and destabilization of these perennially frozen soils (Jorgenson et al., 2006), rendering this vast, aged carbon stock vulnerable to change (Schuur et al., 2015). Permafrost thaw exposes any associated organic matter to mechanical and thermal erosion and can promote microbial decomposition and/or photodegradation (Ward & Cory, 2016) that subsequently leads to the release of CO_2 or methane (Schuur et al., 2008). In the last 50 years, mean annual air temperatures in northern Canada have increased by 2.3°C (and winter temperatures by 4.3°C), roughly 2–3 times the global average (Bush & Lemmen, 2019). Estimates of the impact of permafrost degradation on high-latitude soil carbon stocks predict a loss of between ~10 and ~170 PgC by the end of this century (Koven et al., 2015; Schaefer et al., 2014; Schuur et al., 2015), potentially adding to atmospheric



Writing – review & editing: Melissa S. Schwab, Robert G. Hilton, Timothy I. Eglinton, Peter A. Raymond, Suzanne E. Tank, Edward T. Tipper

greenhouse gas burdens and exacerbating future climate change. Given these projections, tracking the fate of this aged carbon pool, and constraining its response to both gradual and abrupt warming and permafrost thaw, remains a priority (Turetsky et al., 2019).

Along with direct release to the atmosphere, aged carbon mobilized during permafrost thaw can enter surface waters (Plaza et al., 2019) in the form of dissolved organic carbon (DOC) (operationally defined as <0.2 to $<0.7 \mu m$), while particulate organic carbon (POC) is largely supplied to freshwater systems by thermokarst and bank erosion (Jorgenson et al., 2006; Vonk et al., 2015). The degradation of permafrost can increase the thickness of the active layer and result in the loss of ground ice, leading to the formation of thaw slumps, taliks, and thermokarst wetlands (Vonk et al., 2015). Deeper groundwater flow paths contribute to the development of supra-permafrost taliks (areas of unfrozen ground) and connect laterally to the drainage network (Woo, 2012). In early spring, the infiltration of meltwater into permafrost soils can activate hydrologically connected lateral perennial thaw zones (Connon et al., 2018; Walvoord et al., 2019). The flow through supra-permafrost taliks may mobilize DOC from previously frozen soils, well before seasonal permafrost begins to thaw (Walvoord et al., 2019). The sources and flux of DOC in Arctic rivers, which has garnered interest due to its potential reactivity (Cory et al., 2013; Holmes et al., 2008; Mann et al., 2015; Vonk et al., 2013) and its intrinsic link to hydrological pathways, may be modified by such ground thaw (Fouché et al., 2020; Spencer et al., 2015). As water flow accesses deeper layers, a decrease in ¹⁴C activity of DOC has been projected (Frey & McClelland, 2009; Schuur et al., 2015; Tank et al., 2012). Old DOC has previously been identified at small scales, for example, in permafrost seeps and small streams (Dean et al., 2018; Mann et al., 2015; Neff et al., 2006; Vonk et al., 2013). However, in the large Arctic rivers (Yenisey, Lena, Ob', Mackenzie, Yukon, Kolyma) the ¹⁴C activity of DOC has been shown to contain "bomb" carbon stemming from atmospheric nuclear weapons testing in the middle-twentieth Century, implying that young and rapidly cycling organic matter-likely derived from surface vegetation and soils-dominates DOC in large rivers (Raymond et al., 2007; Spencer et al., 2015).

In addition to the mobilization of aged carbon pools, permafrost thaw can result in enhanced weathering of mineral soils and bedrock, further modifying riverine aquatic chemistry (e.g., Frey & McClelland, 2009; Kokelj & Burn, 2005; O'Donnell et al., 2016; Tank et al., 2016; Vonk et al., 2015). As such, the export of major ions (e.g., Ca^{2+} and SO_4^{2-}) which can be enriched in deeper mineral soils is expected to increase in all circumpolar rivers as permafrost degradation persists (Frey & McClelland, 2009; Tank et al., 2012, 2016). In contrast, dissolved nitrogen in the form of nitrate, NO_3^- , is bioactive and influenced by cycling through vegetation and microbial respiration. Elevated riverine NO_3^- concentrations have been linked to thaw slumps and thermokarst gullies (Bowden et al., 2008; Harms et al., 2014), as well as enhanced microbial nitrification and subsequent leaching (Fouché et al., 2017; Louiseize et al., 2014). As such, these dissolved inorganic ions can provide additional insight on sources and pathways of the products of permafrost thaw.

The Mackenzie River $(1.8 \times 10^6 \text{ km}^2)$ is a major carbon source to the Arctic Ocean (Hilton et al., 2015) and has experienced increasing DOC fluxes over the last four decades (Tank et al., 2016). About 50% of the basin lies within continuous and discontinuous permafrost zones (Obu et al., 2019), with its two large, northern tributaries (Arctic Red: $21.8 \times 10^3 \text{ km}^2$; Peel River: $70.6 \times 10^3 \text{ km}^2$) draining almost exclusively continuous permafrost (Figure 1a). The northern part of the basin has been particularly affected by climate change over the instrumental record and has been identified as a region whose carbon stocks are under threat from permafrost thaw (Bush & Lemmen, 2019; Kokelj et al., 2017). Here, we explore the hypothesis that enhanced regional warming could modify transport pathways and the contribution of carbon stored in permafrost soils. We investigate the ¹⁴C age of DOC and the flux of major ions carried by the Mackenzie River and its northern tributaries to identify potential processes impacting the export of DOC in a high-latitude river.

2. Methods

We collected river water in May–June 2017–2019, shortly after ice breakup at the high/receding water stage (Figure 1c), for the Mackenzie River at Tsiigehtchic, in the Delta, and for the Peel and Arctic Red Rivers (Figure 1a). In order to assess vertical variation, we used a modified horizontally mounted ~5.1 L Niskin bottle to recover water from different depths (Hilton et al., 2015). River water was transferred to sterilized plastic bags, weighed, stored in the dark, and filtered within 48 hr using precleaned filtration units (2017–2018) or





Figure 1. (a) River depth profile sampling locations along the Mackenzie River (circle) in the Delta (dark gray) and at Tsiigehtchic (light gray) complemented by the major tributaries, the Arctic Red (light blue square) and the Peel River (dark blue diamond). (b) Mackenzie Delta closeup. (c) Discharge at the Mackenzie River at Tsiigehtchic from 2003 to 2019 (http://www.wateroffice.ec.gc.ca). Dots denote sampling dates. (d) DOC concentrations as function of discharge. Shapes denote different sampling periods.

Teflon-lined steel units (2019) through polysethersulfone filters (PES; Ø 142 mm, 0.22 μ m). Filtered water was collected for DOC in 120 ml pre-combusted (450°C, 6 hr) amber bottles, acidified to pH ~2 with 85% H₃PO₄ (120 μ l), and stored cooled and in the dark. Aliquots for ion analysis were collected in acid-washed high density polyethylene (HDPE) bottles. Sediment-laden filters were folded, carefully placed in pre-combusted aluminum foil envelops and immediately frozen.

Major ions were measured at Durham University and ETH Zurich using a Dionex Ion Chromatography system (DX-120, Thermoscientific) with an analytical reproducibility of 5%. In order to help track water pathways, we utilize ratios of $[Ca^{2+}]/[Cl^{-}]$ and $[NO_{3}^{--}]/[Ca^{2+}]$. Cl^{-} is considered as relatively conservative ion in aquatic environments, with minor engagement in biological and geochemical processes and sourced mainly from rainwater (Gaillardet et al., 1999). Dissolved Ca^{2+} is mostly derived from chemical weathering of carbonate minerals in the Mackenzie River basin (Gaillardet et al., 1999; Millot et al., 2003; Tank et al., 2012). As such, $[Ca^{2+}]/[Cl^{-}]$ ratios can be used to compare the relative mobilization of chemical weathering products between sampling periods. In contrast, NO_{3}^{-} is strongly linked to nutrient cycling in organic matter in near-surface soils. Microbial mineralization of organic matter accumulates NO_{3}^{-} during the winter months when the uptake by plants is diminished (Edwards & Jefferies, 2013; Treat et al., 2016). The $[NO_{3}^{-}]/[Ca^{2+}]$ ratio therefore allows us to assess the relative influence of organic matter cycling versus mineral weathering, albeit with the recognition that NO_{3}^{-} is not necessarily conservative and release from

the landscape may not cascade to increases in NO_3^- in river channels (Harms & Jones, 2012; Wickland et al., 2018).

DOC concentration measurements were conducted using a Shimadzu system (TOC-L Series) at the Department of Environmental System Science at ETH Zurich. DOC (10–53 µgC) was converted to CO₂ using a wet chemical oxidation approach (Lang et al., 2012, 2016). Prior to the oxidation inorganic CO₂ was removed by purging. Evolved CO₂ from DOC oxidation was analyzed using a mini carbon dating accelerator mass spectrometer (MICADAS AMS) system equipped with a gas-accepting ion source at the Laboratory for Ion Beam Physics (LIP) at ETH Zurich. Blank assessment was based on the repeated measurements of sucrose (Sigma, $\delta^{13}C = -12.4\%$ VPDB, $F^{14}C = 1.053 \pm 0.003$) and phthalic acid (Sigma, $\delta^{13}C = -33.6\%$ VPDB, $F^{14}C < 0.0025$) standards. The evaluation of constant contamination described in Haghipour et al. (2019) amounted to ~1 µgC.

Supporting information provides details of sample collection and analyses of various carbon species (DOC and POC), nutrients, and major ions retrieved since 2003 near the apex of the Mackenzie Delta (at Tsiigehtchic, upstream of the Arctic Red and Peel inputs) by the Pan-Arctic River Transport of Nutrients, Organic Matter, and Suspended Sediments (PARTNERS) and the Arctic Great Rivers Observatory (ArcticGRO; www.arcticgreatrivers.org) projects.

3. Results

The samples from the Mackenzie River at Tsiigehtchic collected in June 2017, 2018, and 2019 were taken at similar levels of water discharge and similar points on the hydrograph (i.e., shortly after highest peak) to those collected from 2003–2016, although 2018 and 2019 had the lowest June discharge in the data set (Figure 1c). For the Peel River, the ice breakup is typically a few days earlier in May, and the 2017–2019 samples were collected at similar times relative to the discharge peak (Figure S1).

The June 2017, 2018, and 2019 sampling campaigns resulted in similar DOC concentrations for the Mackenzie River at Tsiigehtchic ($6.4 \pm 0.9 \text{ mg/L}$, n = 18; Table S1) (\pm SD), which is within the variability of the average of June samples from 2003 to 2017 from the PARTNERS-ArcticGRO campaign ($5.7 \pm 1.0 \text{ mg/L}$, n = 17) (Figure 1c). The DOC concentration for 2017–2019 samples was also similar for the Mackenzie River in the Delta ($7.3 \pm 0.5 \text{ mg/L}$, n = 22), the Peel River ($3.9 \pm 0.5 \text{ mg/L}$, n = 9), and the Arctic Red River (9.5 ± 2.0 , n = 8).

From 2003–2013 for the Mackenzie River at Tsiigehtchic, DOC-F¹⁴C values in May and June ranged between 1.06 and 0.96, with an average DOC-F¹⁴C value of 1.01 ± 0.04 (n = 10). In 2017, the F¹⁴C values were similar for both sites on the Mackenzie River (Figure 1a and Table S1) and the Arctic Red and Peel Rivers, with an average DOC-F¹⁴C = 1.00 ± 0.03 (n = 22), and showed little variability with depth in the river (Figure S2). In June 2018, for the Mackenzie River at Tsiigehtchic, DOC-F¹⁴C values ranged from 1.02 to 0.73. Aged DOC was present throughout the Mackenzie River system (Figure 2), with an average DOC-F¹⁴C value across all sample of 0.85 ± 0.16 (n = 28; $1,306 \pm 1,530$ ¹⁴C years). In 2018, the DOC was even more ¹⁴C-depleted (values as low as 0.51) in the Arctic Red and Peel Rivers. The Peel and Arctic Red basins drain higher latitudes, with higher proportions of continuous and discontinuous permafrost cover than the Mackenzie River at Tsiigehtchic (Figure 1a). These rivers join the main stem and combine in the Mackenzie Delta (Middle Channel), where old DOC was also prevalent in 2018. It is important to note that the Peel, Arctic Red, and Mackenzie Rivers at Tsiigehtchic drain different basins and collectively encompassing a large drainage area (Figure 1a).

In June 2019, DOC-F¹⁴C values resembled those of years prior to 2017 (average DOC-F¹⁴C = 0.99 ± 0.01 , n = 20). Nonparametric Kruskal-Wallis tests confirm the statistically significant difference between the ¹⁴C content of samples collected in 2018 from the Mackenzie River at Tsiigehtchic and the Mackenzie Delta compared to those sampled in 2003–2017 and in 2019 (Table S2).

Nitrate concentrations ($[NO_3^-]$) in the Mackenzie River at Tsiigehtchic ($1.53 \pm 0.6 \mu \text{mol}$, n = 7) were lower during the spring freshet in 2017 than the long-term observations of the PARTNERS-ArcticGRO projects ($3.9 \pm 1.5 \mu \text{mol}$, n = 17). Similar [NO_3^-] are observed in Arctic Red ($1.8 \pm 1 \mu \text{mol}$, n = 3) and Mackenzie Delta ($2.4 \pm 1.1 \mu \text{mol}$, n = 18) waters, while the Peel River supplied $4.9 \pm 1.2 \mu \text{mol}$ (n = 5) to the Delta in





Figure 2. Radiocarbon activity, expressed as fraction modern ($F^{14}C$), of DOC over time in the Mackenzie River: (a) $F^{14}C$ of DOC. Circles are samples from the Mackenzie River at Tsiigehtchic and the main channel in the Delta, with filled circles samples collected in May and June. Samples from the Arctic Red and Peel Rivers are also shown for 2017–2019. Data from 2004 to 2013 are provided by the ArcticGRO database. The atmospheric bomb ^{14}C curve is shown as a light gray line (Hammer & Levin, 2017). The error bar in the top left corner represents the mean

standard deviation of 14 C measurements. (b) Boxplot of F^{14} C for different years as a function of sampling location, with the median (black line), first and third quartiles (box), and confidence interval (lines) shown.

2017 and 2019. In 2018, we observe a twofold to threefold increase in the $[NO_3^-]$ in the Mackenzie River and all subcatchments (Figure 3 and Table S3). During the spring freshet, we note a significant negative correlation between log-transformed $[NO_3^-]$ and DOC-F¹⁴C (r = -0.54, n = 60, p < 0.001).

4. Discussion

Prior to this study, reported DOC- F^{14} C values in the large Arctic Rivers (Yenisey, Lena, Ob', Yukon, Kolyma) ranged from 0.82 to 1.16 (Barnes et al., 2018), comparable to those of the Mackenzie River up to 2017 (DOC- F^{14} C: 0.80–1.1; Figure 2) and similar to atmospheric 14 CO₂ (Dean et al., 2020; Raymond et al., 2007; Spencer et al., 2015). This suggests that the DOC pool derives from recently formed biomass, for example, plants, surficial soils, and aquatic productivity (Raymond & Bauer, 2001). Aged DOC has been detected in permafrost seeps and small headwater streams in the Arctic (Dean et al., 2018; Mann et al., 2015; Neff et al., 2006; Vonk et al., 2013). However, aged DOC has not previously been documented in larger Arctic rivers, implying efficient removal during riverine transfer (Holmes et al., 2008; Mann et al., 2015; Vonk et al., 2013) or dilution by modern DOC (Dean et al., 2018). Moderately aged DOC from large Arctic rivers has been attributed to the maximum expansion of the active layer and a deeper penetration of groundwater (Barnes et al., 2018). The presence of aged DOC has been observed in low latitude (Moore et al., 2013) and



Figure 3. (a) Dissolved inorganic species in Mackenzie River at Tsiigehtchic. Relationship between $[Ca^{2+}]/[Cl^{-}]$ and $[NO_{3-}]/[Ca^{2+}]$. Circles represent the values from 2003–2016 (ArcticGRO database), squares 2017, diamonds 2018, and triangles 2019. Symbols are color coded for discharge (m³/s). (b) Radiocarbon activity (expressed as F¹⁴C) of DOC versus NO₃⁻ (µmol) during the spring freshet (May and June). The Mackenzie River samples at Tsiigehtchic and the Delta are indicated in gray and black, whereas the Arctic Red and Peel Rivers are depicted in light and dark blue, respectively. Open symbols represent NO₃⁻ values integrated over depth.



temperate watersheds (Evans et al., 2014) and linked to anthropogenic disturbance (Butman et al., 2015; Drake et al., 2019; Griffith et al., 2009). In the context of this prior work, we discuss mechanisms and their drivers that can explain two important features of the decadal time series from the Mackenzie River: (i) the ¹⁴C depletion of DOC in June 2018 across three separate catchment areas and (ii) the return to ¹⁴C-enriched DOC in June 2019.

4.1. The Timing of Sampling

DOC mobilization and supply to rivers can vary as a function of precipitation (Raymond, 2005; Stubbins et al., 2012) and water discharge (Masiello & Druffel, 2001; Tittel et al., 2015). Broad-scale changes in hydrology associated with a shift in DOC source and flux at the time of sampling might explain the June 2018 aged DOC signal. The Mackenzie River hydrograph is regulated by the onset of thaw and has a remarkably consistent annual pattern (Figure 1c). Snow and ice melt induce a rapid increase in streamflow in spring (freshet) which gradually subsides over subsequent months. During the spring freshet, the concentration, composition, and average age of DOC could vary with discharge, perhaps in response to a "pulse-shunt"-like behavior (Raymond et al., 2016). Snowmelt triggers the pulse-like mobilization and transport of DOC and other nutrients to streams and rivers. Associated high-water velocities shunt solutes rapidly downstream, reducing their residence time in the fluvial network and hence the potential for microbial or photochemical remineralization.

However, no relationship between water discharge and DOC-F¹⁴C emerges in the 2003–2017 and 2019 data from the Mackenzie River at Tsiigehtchic. Instantaneous DOC fluxes measured in June 2018 are similar to those in June 2017 and 2019, with differences in flux mainly controlled by the water yield (Figures S3 and S4). DOC concentrations are similar, both as a function of water column depth and over the studied time period (Figure S5). It is therefore difficult to attribute the June 2018¹⁴C depletion of DOC to a feature of the typical annual hydrograph that has been missed in past sampling. When we consider the aged DOC input in the Peel and Arctic Red Rivers (which drain large, separate watersheds), it suggests that any change in the routing of DOC in June 2018 must be widespread. While we cannot constrain its duration, the sampling of these different rivers took place over 7 days. More frequent samples collected in May 2018 from the surface of the Arctic Red during the ice break up are generally ¹⁴C enriched (mean F¹⁴C = 0.96 ± 0.01, *n* = 8), but the final sample on 29 May 2019 was more ¹⁴C depleted (F¹⁴C = 0.89 ± 0.01). In June 2019, DOC-F¹⁴C values were similar to the other years in the data set (Figure 2). Overall, although annually collected "snap shot" water samples cannot resolve the duration of this export event, we note that June 2018 samples were not associated with anomalous discharge conditions and were collected during similar conditions as those from 2003 to 2019.

4.2. Aged DOC From Permafrost Soils and Climatic Warming

The large change in DOC-F¹⁴C values in June 2018 can be viewed in the context of DOC transfer from soil to streams and in particular the delivery of previously stored, old DOC and weathering products to the hydrological network. Hydrological networks can access a much larger area than other potential causes of DOC aging, such as anthropogenic activity, erosional slumping, and wildfires (Butman et al., 2015; Gibson et al., 2018; Kettridge et al., 2012; Kokelj et al., 2013, 2015) (see supporting information discussion). The annual summer and freezing temperatures play important roles in the dynamics and thickness of the active layer (Kokelj & Burn, 2005). Air temperature records from Inuvik and Norman Wells during the freezing season (onset and the end of continuous freezing) show temperature increases over the last 76 years (Bush & Lemmen, 2019) (Figures S6, S7, and Table S4) and a pronounced temperature anomaly in winter 2017/2018. The winter 2018/2019 period shows an even higher temperature anomaly. In addition, we note an anomalous warm summer period in 2017 which was followed by colder summer seasons in 2018 and 2019 (Figure S8).

Observations of permafrost temperature in the northern Mackenzie River basin point to a warming of 0.5°C to 0.9°C and a thickening of the active layer by about 10% since 2000 (Biskaborn et al., 2019; Bush & Lemmen, 2019; Smith et al., 2018). The shift of the permafrost table due to thaw during the summer months occurs both vertically and laterally and allows the development of thin, perennial taliks within and above permafrost (Lamontagne-Hallé et al., 2018; Walvoord et al., 2012; Zhang et al., 2008). Mild winters are insufficiently cold to completely counterbalance this thaw. Increased hydrogeologic connectivity in the fall

enhances the drainage of surface soils (Liljedahl et al., 2016), the reduction of soil moisture, and the temporary increase of the water storage capacity (McCartney et al., 2006; Quinton et al., 2005; Teufel & Sushama, 2019). As a consequence, organic soils can be undersaturated prior to freeze back in fall. Under this regime, a larger portion of meltwater can infiltrate soils in spring, rather than entering streams as surface runoff (Teufel & Sushama, 2019). Percolating meltwater supplies sensible heat to the soil leading to the thaw of the upper permafrost table and the expansion of hydrologically connected pathways (Connon et al., 2018; Teufel & Sushama, 2019; Walvoord et al., 2019). Organic-rich material concentrated below the active layer can be mobilized by this subsurface flow of water, mixed with meltwater, and subsequently transferred to streams (Plaza et al., 2019; Walvoord & Striegl, 2007; Walvoord et al., 2019). Supra-permafrost taliks are expected to become more pronounced with continued warming, as a result of delayed active layer freeze up in fall accompanied by a poor freeze back in warm winters (Euskirchen et al., 2006; Serreze et al., 2000; Walvoord et al., 2019).

To better constrain the mechanisms operating in June 2018, we examine dissolved ions that originate from mineral weathering and organic matter cycling. We find that June 2018 waters exhibit higher abundances of calcium ions relative to Cl⁻ ions, a conservative tracer, and enhanced relative abundances of NO₃⁻ ions (Figure 3a). In addition, DOC $F^{14}C$ values were negatively correlated with $[NO_3^{-}]$ at all locations (Figure 3b). An increase input of NO_3^- to high-latitude streams, rivers, and lakes has been associated with gradual warming (e.g., Fouché et al., 2020; Frey et al., 2007; Harms & Jones, 2012; Jones et al., 2005; McClelland et al., 2007; Walvoord & Striegl, 2007), the draining of thermokarst environments (Abbott et al., 2015), and the more frequent occurrence of wildfires (Petrone et al., 2007). However, the fate of solute transfer from permafrost soils to streams depends on abiotic and biotic processes, complex interactions within the soil matrix, and residence times (Harms & Jones, 2012; Spencer et al., 2015; Striegl et al., 2005). While Ca^{2+} ions are derived from carbonate weathering in the Mackenzie River basin (Tank et al., 2016), NO₃⁻ is produced by nitrification following organic matter mineralization and the dieback of plant roots (Keuper et al., 2012; Treat et al., 2016). In fall and winter, NO₃⁻ can accumulate as a result of continued microbial remineralization, while the uptake by plants is reduced due to dormancy (Edwards & Jefferies, 2013; Treat et al., 2016). Extractions performed on permafrost soils show that both Ca^{2+} and NO₃⁻ ions can be immobilized by freezing and enriched relative to other ions just below the active layer (Kokelj & Burn, 2005; Reyes & Lougheed, 2015). The release of aged DOC in 2018 was accompanied by a doubling in [NO₃⁻] (Figure 3b and Table S3). This suggests that waters accessing and routed through soils near the permafrost table serve as a viable source of aged DOC in June 2018.

Although the winter 2018/19 experienced record warmth, we observe no aged DOC in the June 2019 samples. However, preceding summer temperatures in 2018 are lower than the long-term average. Indeed, we find that June 2019 waters have lower $[NO_3^{-}]/[Ca^{2+}]$ ratios and similar $[NO_3^{-}]$ as June samples from 2003–2017 (Figure 3a). The annual thaw depth strongly varies as a function of near-surface temperature (Kokelj & Burn, 2005; Teufel & Sushama, 2019). Given the cooler summer in 2018, it is possible that the annual thaw was less severe and likely supported the formation of soil ice that counteracted the development of taliks and hydraulically connected pathways in the following spring 2019. Another explanation could be that our sampling missed an aged DOC signature that occurred earlier or later in 2019 (Figure 1c). Alternatively, it could reflect an exhaustion of DOC first mobilized in June 2018 and/or concurrent increases in mobilization of young surface organic carbon (Dean et al., 2018; Feng et al., 2013). Differences in the contributions from specific higher-order streams and/or the biodegradability of DOC during transport (Holmes et al., 2008; Mann et al., 2015; Vonk et al., 2013) could have also played a role.

While it is not possible from our data set to unravel the details of carbon mobilization in permafrost zones undergoing thaw, our work highlights the potential transitory nature of aged DOC export events. Despite the challenges of long-term sampling of river carbon species and coupled analyses of ¹⁴C activity, our observations underline the potential for rapid and large-scale mobilization of aged carbon pools previously sequestered in Arctic permafrost soils.

5. Conclusions

In this study, radiocarbon measurements of DOC in the Mackenzie River and its two large northern tributaries (Peel and Arctic Red Rivers) from 2017 to 2019 are combined with previously published data to



explore interannual variability in DOC export from a major Arctic fluvial system. DOC ages from 2003 to 2017 suggest a predominant source from recent vegetation and soils. However, samples from June 2018 record a previously undocumented and widespread export of aged DOC, before returning to modern $F^{14}C$ values in June 2019. This aged DOC was accompanied by high $[NO_3^-]$ in all three rivers. The change in DOC age and solute flux that we observe appears consistent with the formation of supra-permafrost taliks and the deeper percolation of groundwater, resulting in mobilization of organic matter and weathering products preserved in previously frozen, organic-rich soil horizons. Limitations in sampling coverage preclude elucidation of the magnitude and dynamics of aged DOC export and associated carbon cycle feedbacks in response to a warming climate and highlight the need for sustained, long-term observations and refined sampling and measurement strategies. Irrespective of these caveats, our observations reveal the potential for abrupt and pervasive mobilization of hitherto sequestered carbon from vast permafrost regions of the Arctic.

Conflict of Interest

The authors declare no conflict of interest.

Data Availability Statement

Data from 2003 to 2016 are available in the ArcticGRO repository (https://arcticgreatrivers.org/data/). All data generated from 2017 to 2019 are openly available in the EarthChem Library (https://doi.org/ 10.26022/IEDA/111725). A detailed method description and supporting information figures and tables can be found in the supporting information. Correspondence and requests for materials should be addressed to M. S. S. (melissa.schwab@erdw.ethz.ch).

References

- Abbott, B. W., Jones, J. B., Godsey, S. E., Larouche, J. R., & Bowden, W. B. (2015). Patterns and persistence of hydrologic carbon and nutrient export from collapsing upland permafrost. *Biogeosciences*, *12*(12), 3725–3740. https://doi.org/10.5194/bg-12-3725-2015
- Barnes, R. T., Butman, D. E., Wilson, H. F., & Raymond, P. A. (2018). Riverine export of aged carbon driven by flow path depth and residence time. *Environmental Science and Technology*, 52(3), 1028–1035. https://doi.org/10.1021/acs.est.7b04717
 - Biskaborn, B. K., Smith, S. L., Noetzli, J., Matthes, H., Vieira, G., Streletskiy, D. A., et al. (2019). Permafrost is warming at a global scale. *Nature Communications*, 10(1), 1–11. https://doi.org/10.1038/s41467-018-08240-4
 - Bowden, W. B., Gooseff, M. N., Balser, A., Green, A., Peterson, B. J., & Bradford, J. (2008). Sediment and nutrient delivery from thermokarst features in the foothills of the North Slope, Alaska: Potential impacts on headwater stream ecosystems. *Journal of Geophysical Research*, *113*, G02026. https://doi.org/10.1029/2007JG000470
 - Bush, E., & Lemmen, D. S. (Eds.) (2019). Canada's changing climate report (pp. 1–444). Ottawa, ON: Goverment of Canada. https://www. ChangingClimate.ca/CCCR2019
 - Butman, D. E., Wilson, H. F., Barnes, R. T., Xenopoulos, M. A., & Raymond, P. A. (2015). Increased mobilization of aged carbon to rivers by human disturbance. *Nature Geoscience*, 8(2), 112–116. https://doi.org/10.1038/ngeo2322
 - Carson, M. A., Jasper, J. N., & Conly, F. M. (1998). Magnitude and sources of sediment input to the Mackenzie Delta, Northwest Territories. *Arctic*, *51*, 116–124.
 - Connon, R., Devoie, É., Hayashi, M., Veness, T., & Quinton, W. (2018). The influence of shallow taliks on permafrost thaw and active layer dynamics in subarctic Canada. Journal of Geophysical Research: Earth Surface, 123, 281–297. https://doi.org/10.1002/2017JF004469
 - Cory, R. M., Crump, B. C., Dobkowski, J. A., & Kling, G. W. (2013). Surface exposure to sunlight stimulates CO₂ release from permafrost soil carbon in the Arctic. Proceedings of the National Academy of Sciences of the United States of America, 110(9), 3429–3434. https://doi.org/ 10.1073/pnas.1214104110
 - Dean, J. F., Meisel, O. H., Martyn Rosco, M., Marchesini, L. B., Garnett, M. H., Lenderink, H., et al. (2020). East Siberian Arctic inland waters emit mostly contemporary carbon. *Nature Communications*, *11*(1), 1–10. https://doi.org/10.1038/s41467-020-15511-6
 - Dean, J. F., van der Velde, Y., Garnett, M. H., Dinsmore, K. J., Baxter, R., Lessels, J. S., et al. (2018). Abundant pre-industrial carbon detected in Canadian Arctic headwaters: Implications for the permafrost carbon feedback. *Environmental Research Letters*, 13(3), 034024. https://doi.org/10.1088/1748-9326/aaa1fe
 - Drake, T. W., van Oost, K., Barthel, M., Bauters, M., Hoyt, A. M., Podgorski, D. C., et al. (2019). Mobilization of aged and biolabile soil carbon by tropical deforestation. *Nature Geoscience*, 12(July), 541–546. https://doi.org/10.1038/s41561-019-0384-9
 - Edwards, K. A., & Jefferies, R. L. (2013). Inter-annual and seasonal dynamics of soil microbial biomass and nutrients in wet and dry low-Arctic sedge meadows. *Soil Biology and Biochemistry*, 57, 83–90. https://doi.org/10.1016/j.soilbio.2012.07.018
 - Euskirchen, E. S., McGuire, A. D., Kicklighter, D. W., Zhuang, Q., Clein, J. S., Dargaville, R. J., et al. (2006). Importance of recent shifts in soil thermal dynamics on growing season length, productivity, and carbon sequestration in terrestrial high-latitude ecosystems. *Global Change Biology*, *12*(4), 731–750. https://doi.org/10.1111/j.1365-2486.2006.01113.x
 - Evans, C. D., Page, S. E., Jones, T., Moore, S., Gauci, V., Laiho, R., et al. (2014). Contrasting vulnerability of drained tropical and high-latitude peatlands to fluvial loss of stored carbon. *Global Biogeochemical Cycles*, 28, 1215–1234. https://doi.org/10.1002/ 2013GB004782
 - Feng, X., Vonk, J. E., van Dongen, B. E., Gustafsson, O., Semiletov, I. P., Dudarev, O. V., et al. (2013). Differential mobilization of terrestrial carbon pools in Eurasian Arctic river basins. Proceedings of the National Academy of Sciences of the United States of America, 110(35), 14,168–14,173. https://doi.org/10.1073/pnas.1307031110

Acknowledgments

M. S. S. was supported by the Swiss National Science Foundation through the grant SNF200020_163162/1, "CAPS-LOCK II" and SNF200020_184865/1, "CAPS-LOCK III." R. G. H. was funded by the European Research Council (ERC) Starting Grant 678779, "ROC-CO2" and a NERC UK-Canada Arctic Partnership Bursaries Program (R. G. H, S. E. T, and E. T. T.). Additional logistical support was provided by the Aurora Research Institute. We thank Matthieu Dellinger, Christina Larkin, Jotautas Baronas, Gabriela Santilli, Barbara Lesniak, and Daniel Montluçon for assistance with sample collection, Björn Studer for dissolved organic carbon concentration measurements, and Fanny Leuenberger-West and Amanda Hayton for technical support. Supporting information is available for this paper.

- Fouché, J., Christiansen, C. T., Lafrenière, M. J., Grogan, P., & Lamoureux, S. F. (2020). Canadian permafrost stores large pools of ammonium and optically distinct dissolved organic matter. *Nature Communications*, 11(1), 1–11. https://doi.org/10.1038/s41467-020-18331-w
- Fouché, J., Lafrenière, M. J., Rutherford, K., & Lamoureux, S. (2017). Seasonal hydrology and permafrost disturbance impacts on dissolved organic matter composition in High Arctic headwater catchments. *Arctic Science*, 3(2), 378–405. https://doi.org/10.1139/as-2016-0031 Frey, K. E., & McClelland, J. W. (2009). Impacts of permafrost degradation on arctic river biogeochemistry. *Hydrological Processes*, 23,
- Frey, K. E., & McClelland, J. W. (2009). Impacts of permainost degradation on arctic river biogeochemistry. *Hydrological Processes*, 23, 169–182. https://doi.org/10.1002/hyp
 Frey, K. E., McClelland, J. W., Holmes, R. M., & Smith, L. G. (2007). Impacts of climate warming and permafrost thaw on the riverine
- Frey, K. E., McClenand, J. W., Holmes, K. M., & Smith, L. G. (2007). Impacts of climate warming and permainost thaw on the riverine transport of nitrogen and phosphorus to the Kara Sea. *Journal of Geophysical Research*, 112, G04S58. https://doi.org/10.1029/ 2006JG000369
- Gaillardet, J., Dupre, B., Louvat, P., & Allegre, C. J. (1999). Global silicate weathering and CO₂ consumption rates deduced from the chemistry of large rivers. *Chemical Geology*, 159, 3–30. https://doi.org/10.1016/S0009-2541(99)00031-5
- Gibson, C. M., Chasmer, L. E., Thompson, D. K., Quinton, W. L., Flannigan, M. D., & Olefeldt, D. (2018). Wildfire as a major driver of recent permafrost thaw in boreal peatlands. *Nature Communications*, 9(1), 3041. https://doi.org/10.1038/s41467-018-05457-1
- Griffith, D. R., Barnes, R. T., & Raymond, P. A. (2009). Inputs of fossil carbon from wastewater treatment plants to U.S. Rivers and oceans. Environmental Science and Technology, 43(15), 5647–5651. https://doi.org/10.1021/es9004043
- Haghipour, N., Ausin, B., Usman, M. O., Ishikawa, N., Wacker, L., Welte, C., et al. (2019). Compound-specific radiocarbon analysis by elemental analyzer-accelerator mass spectrometry: Precision and limitations. *Analytical Chemistry*, 91(3), 2042–2049. https://doi.org/ 10.1021/acs.analchem.8b04491
- Hammer, S., & Levin, I. (2017). Monthly mean atmospheric Δ^{14} CO₂ at Jungfraujoch and Schauinsland from 1986 to 2016. heiDATA, V2. Retrieved from https://doi.org/10.11588/data/10100
- Harms, T. K., Abbott, B. W., & Jones, J. B. (2014). Thermo-erosion gullies increase nitrogen available for hydrologic export. *Biogeochemistry*, 117(2–3), 299–311. https://doi.org/10.1007/s10533-013-9862-0
- Harms, T. K., & Jones, J. B. (2012). Thaw depth determines reaction and transport of inorganic nitrogen in valley bottom permafrost soils. *Global Change Biology*, 18(9), 2958–2968. https://doi.org/10.1111/j.1365-2486.2012.02731.x
- Hilton, R. G., Galy, V., Gaillardet, J., Dellinger, M., Bryant, C., O'Regan, M., et al. (2015). Erosion of organic carbon in the Arctic as a geological carbon dioxide sink. *Nature*, 524(7563), 84–87. https://doi.org/10.1038/nature14653
- Holmes, R. M., McClelland, J. W., Raymond, P. A., Frazer, B. B., Peterson, B. J., & Stieglitz, M. (2008). Lability of DOC transported by Alaskan rivers to the Arctic Ocean. *Geophysical Research Letters*, 35, L03402. https://doi.org/10.1029/2007GL032837
- Hugelius, G., Strauss, J., Zubrzycki, S., Harden, J. W., Schuur, E. A. G., Ping, C. L., et al. (2014). Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. *Biogeosciences*, 11(23), 6573–6593. https://doi.org/10.5194/bg-11-6573-2014
- Jones, J. B., Petrone, K. C., Finlay, J. C., Hinzman, L. D., & Bolton, W. R. (2005). Nitrogen loss from watersheds of interior Alaska underlain with discontinuous permafrost. *Geophysical Research Letters*, 32, L02401. https://doi.org/10.1029/2004GL021734
- Jorgenson, M. T., Shur, Y. L., & Pullman, E. R. (2006). Abrupt increase in permafrost degradation in Arctic Alaska. Geophysical Research Letters, 33, L02503. https://doi.org/10.1029/2005GL024960
- Kettridge, N., Thompson, D. K., & Waddington, J. M. (2012). Impact of wildfire on the thermal behavior of northern peatlands: Observations and model simulations. *Journal of Geophysical Research*, *117*, G02014. https://doi.org/10.1029/2011JG001910
- Keuper, F., van Bodegom, P. M., Dorrepaal, E., Weedon, J. T., van Hal, J., van Logtestijn, R. S. P., & Aerts, R. (2012). A frozen feast: Thawing permafrost increases plant-available nitrogen in subarctic peatlands. *Global Change Biology*, 18(6), 1998–2007. https://doi.org/10.1111/ j.1365-2486.2012.02663.x
- Kokelj, S. V., & Burn, C. R. (2005). Geochemistry of the active layer and near-surface permafrost, Mackenzie delta region, Northwest Territories, Canada. Canadian Journal of Earth Sciences, 42(1), 37–48. https://doi.org/10.1139/e04-089
- Kokelj, S. V., Lacelle, D., Lantz, T. C., Tunnicliffe, J., Malone, L., Clark, I. D., & Chin, K. S. (2013). Thawing of massive ground ice in mega slumps drives increases in stream sediment and solute flux across a range of watershed scales. *Journal of Geophysical Research: Earth* Surface, 118, 681–692. https://doi.org/10.1002/jgrf.20063
- Kokelj, S. V., Lantz, T. C., Tunnicliffe, J., Segal, R., & Lacelle, D. (2017). Climate-driven thaw of permafrost preserved glacial landscapes, northwestern Canada. *Geology*, 45(4), 371–374. https://doi.org/10.1130/G38626.1
- Kokelj, S. V., Tunnicliffe, J., Lacelle, D., Lantz, T. C., Chin, K. S., & Fraser, R. (2015). Increased precipitation drives mega slump development and destabilization of ice-rich permafrost terrain, northwestern Canada. *Global and Planetary Change*, 129, 56–68. https://doi. org/10.1016/j.gloplacha.2015.02.008
- Koven, C. D., Schuur, E. A. G., Schädel, C., Bohn, T. J., Burke, E. J., Chen, G., et al. (2015). A simplified, data-constrained approach to estimate the permafrost carbon-climate feedback. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 373(2054), 20140423. https://doi.org/10.1098/rsta.2014.0423
- Lamontagne-Hallé, P., McKenzie, J. M., Kurylyk, B. L., & Zipper, S. C. (2018). Changing groundwater discharge dynamics in permafrost regions. *Environmental Research Letters*, 13(8). https://doi.org/10.1088/1748-9326/aad404
- Lang, S. Q., Bernasconi, S. M., & Früh-Green, G. L. (2012). Stable isotope analysis of organic carbon in small (μg C) samples and dissolved organic matter using a GasBench preparation device. *Rapid Communications in Mass Spectrometry*, *26*(1), 9–16. https://doi.org/10.1002/ rcm.5287
- Lang, S. Q., Mcintyre, C. P., Bernasconi, S. M., Früh-Green, G. L., Voss, B. M., Eglinton, T. I., & Wacker, L. (2016). Rapid ¹⁴C analysis of dissolved organic carbon in non-saline waters. *Radiocarbon*, 58(3), 1–11. https://doi.org/10.1017/RDC.2016.17
- Liljedahl, A. K., Boike, J., Daanen, R. P., Fedorov, A. N., Frost, G. V., Grosse, G., et al. (2016). Pan-Arctic ice-wedge degradation in warming permafrost and its influence on tundra hydrology. *Nature Geoscience*, 9(4), 312–318. https://doi.org/10.1038/ngeo2674
- Louiseize, N. L., Lafrenière, M. J., & Hastings, M. G. (2014). Stable isotopic evidence of enhanced export of microbially derived NO₃⁻ following active layer slope disturbance in the Canadian High Arctic. *Biogeochemistry*, *121*(3), 565–580. https://doi.org/10.1007/s10533-014-0023-x
- Mann, P. J., Eglinton, T. I., McIntyre, C. P., Zimov, N., Davydova, A., Vonk, J. E., et al. (2015). Utilization of ancient permafrost carbon in headwaters of Arctic fluvial networks. *Nature Communications*, 6(1), 7856–7857. https://doi.org/10.1038/ncomms8856
- Masiello, A., & Druffel, E. R. M. (2001). Carbon isotope geochemistry of the Santa Clara River. *Global Biogeochemical Cycles*, *15*(2), 407–416.
 McCartney, S. E., Carey, S. K., & Pomeroy, J. W. (2006). Intra-basin variability of snowmelt water balance calculations in a subarctic catchment. *Hydrological Processes*, *20*(4), 1001–1016. https://doi.org/10.1002/hyp.6125

- McClelland, J. W., Stieglitz, M., Pan, F., Holmes, R. M., & Peterson, B. J. (2007). Recent changes in nitrate and dissolved organic carbon export from the upper Kuparuk River, North Slope, Alaska. *Journal of Geophysical Research*, 112, G04S60. https://doi.org/10.1029/ 2006JG000371
- Millot, R., Gaillardet, J., Dupré, B., & Allégre, C. J. (2003). Northern latitude chemical weathering rates: Clues from the Mackenzie River Basin, Canada. Geochimica et Cosmochimica Acta, 67(7), 1305–1329. https://doi.org/10.1016/S0016-7037(02)01207-3
- Moore, S., Evans, C. D., Page, S. E., Garnett, M. H., Jones, T. G., Freeman, C., et al. (2013). Deep instability of deforested tropical peatlands revealed by fluvial organic carbon fluxes. *Nature*, 493(7434), 660–663. https://doi.org/10.1038/nature11818
- Neff, J. C., Finlay, J. C., Zimov, S. A., Davydov, S. P., Carrasco, J. J., Schuur, E. A. G., & Davydova, A. I. (2006). Seasonal changes in the age and structure of dissolved organic carbon in Siberian rivers and streams. *Geophysical Research Letters*, 33, L23401. https://doi.org/ 10.1029/2006GL028222
- Obu, J., Westermann, S., Bartsch, A., Berdnikov, N., Christiansen, H. H., Dashtseren, A., et al. (2019). Northern Hemisphere permafrost map based on TTOP modelling for 2000–2016 at 1 km² scale. *Earth-Science Reviews*, 193(October 2018), 299–316. https://doi.org/ 10.1016/j.earscirev.2019.04.023
- O'Donnell, J. A., Aiken, G. R., Swanson, D. K., Panda, S., Butler, K. D., & Baltensperger, A. P. (2016). Dissolved organic matter composition of Arctic rivers: Linking permafrost and parent material to riverine carbon. *Global Biogeochemical Cycles*, 30, 1811–1826. https://doi.org/ 10.1002/2016GB005482
- Petrone, K. C., Hinzman, L. D., Shibata, H., Jones, J. B., & Boone, R. D. (2007). The influence of fire and permafrost on sub-arctic stream chemistry during storms. *Hydrolical Processes*, *21*, 423–434. https://doi.org/10.1002/hyp
- Ping, C. L., Michaelson, G. J., Guo, L., Jorgenson, M. T., Kanevskiy, M., Shur, Y., et al. (2011). Soil carbon and material fluxes across the eroding Alaska Beaufort Sea coastline. *Journal of Geophysical Research*, 116, G02004. https://doi.org/10.1029/2010JG001588
- Plaza, C., Pegoraro, E., Bracho, R., Celis, G., Crummer, K. G., Hutchings, J. A., et al. (2019). Direct observation of permafrost degradation and rapid soil carbon loss in tundra. *Nature Geoscience*, *12*(August), 627–631. https://doi.org/10.1038/s41561-019-0387-6
- Quinton, W. L., Shirazi, T., Carey, S. K., & Pomeroy, J. W. (2005). Soil water storage and active-layer development in a Sub-Alpine Tundra hillslope, southern Yukon Territory, Canada. *Permafrost and Periglacial Processes*, 16(4), 369–382. https://doi.org/10.1002/ ppp.543
- Raymond, P. A. (2005). The composition and transport of organic carbon in rainfall: Insights from the natural (¹³C and ¹⁴C) isotopes of carbon. *Geophysical Research Letters*, *32*, L14402. https://doi.org/10.1029/2005GL022879
- Raymond, P. A., & Bauer, J. E. (2001). Use of ¹⁴C and ¹³C natural abundances for evaluating riverine, estuarine, and coastal DOC and POC sources and cycling: A review and synthesis: Organic geochemical tracers in estuaries. Organic Geochemistry, 32(4), 469–485.
- Raymond, P. A., McClelland, J. W., Holmes, R. M., Zhulidov, A. V., Mull, K., Peterson, B. J., et al. (2007). Flux and age of dissolved organic carbon exported to the Arctic Ocean: A carbon isotopic study of the five largest arctic rivers. *Global Biogeochemical Cycles*, 21, GB4011. https://doi.org/10.1029/2007GB002934

Raymond, P. A., Saiers, J. E., & Sobczak, W. V. (2016). Hydrological and biogeochemical controls on watershed DOM transport: Pulse-shunt concept. *Ecology*, 97(1), 5–16. https://doi.org/10.1890/14-1684.1

Reyes, F. R., & Lougheed, V. L. (2015). Rapid nutrient release from permafrost thaw in Arctic aquatic ecosystems. Arctic, Antarctic, and Alpine Research, 47(1), 35–48. https://doi.org/10.1657/aaar0013-099

- Schaefer, K., Lantuit, H., Romanovsky, V. E., Schuur, E. A. G., & Witt, R. (2014). The impact of the permafrost carbon feedback on global climate. *Environmental Research Letters*, 9(8). https://doi.org/10.1088/1748-9326/9/8/085003
- Schuur, E. A. G., Bockheim, J., & Canadell, J. (2008). Vulnerability of permafrost carbon to climate change: Implications for the global carbon cycle. *Bioscience*, 58(8), 701–714. https://doi.org/10.1641/B580807
- Schuur, E. A. G., McGuire, A. D., Schädel, C., Grosse, G., Harden, J. W., Hayes, D. J., et al. (2015). Climate change and the permafrost carbon feedback. *Nature*, 520(7546), 171–179. https://doi.org/10.1038/nature14338
- Serreze, M. C., Walsh, J. E., Chapin, F. S. III, Osterkamp, T., Dyurgerov, M., Romanovsky, V., et al. (2000). Observational evidence of recent change in the northern high-latitude environment. *Climatic Change*, 46(1/2), 159–207. https://doi.org/10.1023/A:1005504031923
- Smith, S. L., Chartrand, J., Duchesne, C., Smith, S. L., Chartrand, J., & Duchesne, C. (2018). Report on 2017 field activities and collection of ground-thermal and active-layer data in the Mackenzie corridor, Northwest Territories; Geological Survey of Canada, Open file 8492.
- Spencer, R. G., Mann, P. J., Dittmar, T., Eglinton, T. I., McIntyre, C., Holmes, R. M., et al. (2015). Detecting the signature of permafrost thaw in Arctic rivers. *Geophysical Research Letters*, 42, 2830–2835. https://doi.org/10.1002/2015GL063498
- Striegl, R. G., Aiken, G. R., Dornblaser, M. M., Raymond, P. A., & Wickland, K. P. (2005). A decrease in discharge-normalized DOC export by the Yukon River during summer through autumn. *Geophysical Research Letters*, 32, L21413. https://doi.org/10.1029/ 2005GL024413
- Stubbins, A., Hood, E., Raymond, P. A., Aiken, G. R., Sleighter, R. L., Hernes, P. J., et al. (2012). Anthropogenic aerosols as a source of ancient dissolved organic matter in glaciers. *Nature Geoscience*, 5(3), 198–201. https://doi.org/10.1038/ngeo1403
- Tank, S. E., Raymond, P. A., Striegl, R. G., McClelland, J. W., Holmes, R. M., Fiske, G. J., & Peterson, B. J. (2012). A land-to-ocean perspective on the magnitude, source and implication of DIC flux from major Arctic rivers to the Arctic Ocean. *Global Biogeochemical Cycles*, 26, GB4018. https://doi.org/10.1029/2011GB004192
- Tank, S. E., Striegl, R. G., McClelland, J. W., & Kokelj, S. V. (2016). Multi-decadal increases in dissolved organic carbon and alkalinity flux from the Mackenzie drainage basin to the Arctic Ocean. *Environmental Research Letters*, *11*(5), 054015. Retrieved from. http://stacks.iop. org/1748-9326/11/i=5/a=054015
- Teufel, B., & Sushama, L. (2019). Abrupt changes across the Arctic permafrost region endanger northern development. *Nature Climate Change*, 9(11), 858–862. https://doi.org/10.1038/s41558-019-0614-6
- Tittel, J., Müller, C., Schultze, M., Musolff, A., & Knöller, K. (2015). Fluvial radiocarbon and its temporal variability during contrasting hydrological conditions. *Biogeochemistry*, 126(1–2), 57–69. https://doi.org/10.1007/s10533-015-0137-9
- Treat, C. C., Wollheim, W. M., Varner, R. K., & Bowden, W. B. (2016). Longer thaw seasons increase nitrogen availability for leaching during fall in tundra soils. *Environmental Research Letters*, 11(6). https://doi.org/10.1088/1748-9326/11/6/064013
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Walter Anthony, K., Olefeldt, D., Schuur, E. A. G., et al. (2019). Permafrost collapse is accelerating carbon release. *Nature*, 569(7754), 32–34. https://doi.org/10.1038/d41586-019-01313-4
- Vonk, J. E., Mann, P. J., Davydov, S., Davydova, A., Spencer, R. G. M., Schade, J., et al. (2013). High biolability of ancient permafrost carbon upon thaw. *Geophysical Research Letters*, 40, 2689–2693. https://doi.org/10.1002/grl.50348
- Vonk, J. E., Tank, S. E., Bowden, W. B., Laurion, I., Vincent, W. F., Alekseychik, P., et al. (2015). Reviews and syntheses: Effects of permafrost thaw on Arctic aquatic ecosystems. *Biogeosciences*, 12(23), 7129–7167. https://doi.org/10.5194/bg-12-7129-2015



- Walvoord, M. A., & Striegl, R. G. (2007). Increased groundwater to stream discharge from permafrost thawing in the Yukon River basin: Potential impacts on lateral export of carbon and nitrogen. *Geophysical Research Letters*, 34, L12402. https://doi.org/10.1029/ 2007GL030216
- Walvoord, M. A., Voss, C. I., Ebel, B. A., & Minsley, B. J. (2019). Development of perennial thaw zones in boreal hillslopes enhances potential mobilization of permafrost carbon. *Environmental Research Letters*, 14(1). https://doi.org/10.1088/1748-9326/aaf0cc
- Walvoord, M. A., Voss, C. I., & Wellman, T. P. (2012). Influence of permafrost distribution on groundwater flow in the context of climate-driven permafrost thaw: Example from Yukon Flats Basin, Alaska, United States. *Water Resources Research*, 48, W07524. https://doi.org/10.1029/2011WR011595
- Ward, C. P., & Cory, R. M. (2016). Complete and partial photo-oxidation of dissolved organic matter draining permafrost soils. Environmental Science and Technology, 50(7), 3545–3553. https://doi.org/10.1021/acs.est.5b05354
- Wickland, K. P., Waldrop, M. P., Aiken, G. R., Koch, J. C., Jorgenson, M. T., & Striegl, R. G. (2018). Dissolved organic carbon and nitrogen release from boreal Holocene permafrost and seasonally frozen soils of Alaska. *Environmental Research Letters*, 13(6). https://doi.org/ 10.1088/1748-9326/aac4ad
- Woo, M. K. (2012). Permafrost hydrology. Permafrost hydrology (pp. 1–564). Berlin, Heidelberg: Springer Berlin Heidelberg. https://doi.org/ 10.1007/978-3-642-23462-0
- Zhang, Y., Chen, W., & Riseborough, D. W. (2008). Disequilibrium response of permafrost thaw to climate warming in Canada over 1850-2100. Geophysical Research Letters, 35, L02502. https://doi.org/10.1029/2007GL032117
- Zimov, S. A., Schuur, E. A. G., & Stuart Chapin, F. III (2006). Permafrost and the global carbon budget. *Science*, 312(5780), 1612–1613. https://doi.org/10.1126/science.1128908

References From the Supporting Information

- Bao, R., McNichol, A. P., McIntyre, C. P., Xu, L., & Eglinton, T. I. (2018). Dimensions of radiocarbon variability within sedimentary organic matter. Radiocarbon, 60(3), 775–790. https://doi.org/10.1017/RDC.2018.22
- Brown, D., Jorgenson, M. T., Douglas, T., & Ruess, R. (2015). Interactions of fire and climate exacerbate permafrost degradation in Alaskan lowland forests. *Journal of Geophysical Research: Biogeosciences*, 120, 1619–1637. https://doi.org/10.1002/2015JG003033
- Flannigan, M., Stocks, B., Turetsky, M., & Wotton, M. (2009). Impacts of climate change on fire activity and fire management in the circumboreal forest. *Global Change Biology*, 15(3), 549–560. https://doi.org/10.1111/j.1365-2486.2008.01660.x
- Goñi, M. A., Hatten, J. A., Wheatcroft, R. A., & Borgeld, J. C. (2013). Particulate organic matter export by two contrasting small mountainous rivers from the Pacific Northwest, U. S.A. Journal of Geophysical Research: Biogeosciences, 118, 112–134. https://doi.org/10.1002/ jgrg.20024
- Goñi, M. A., Yunker, M. B., Macdonald, R. W., & Eglinton, T. I. (2005). The supply and preservation of ancient and modern components of organic carbon in the Canadian Beaufort Shelf of the Arctic Ocean. *Marine Chemistry*, 93(1), 53–73. https://doi.org/10.1016/j. marchem.2004.08.001
- Hein, F. J. (2006). Heavy oil and oil (tar) sands in North America: An overview & summary of contributions. *Natural Resources Research*, 15(2), 67–84. https://doi.org/10.1007/s11053-006-9016-3
- Hinzman, L. D., Fukuda, M., Sandberg, D. V., Chapin, F. S., & Dash, D. (2003). FROSTFIRE: An experimental approach to predicting the climate feedbacks from the changing boreal fire regime. *Journal of Geophysical Research*, 108(D1), 8153. https://doi.org/10.1029/ 2001JD000415
- Holmes, R. M., McClelland, J. W., Peterson, B. J., Tank, S. E., Bulygina, E., Eglinton, T. I., et al. (2012). Seasonal and annual fluxes of nutrients and organic matter from large rivers to the Arctic Ocean and surrounding seas. *Estuaries and Coasts*, 35(2), 369–382. https:// doi.org/10.1007/s12237-011-9386-6
- Hood, E., Fellman, J., Spencer, R. G. M., Hernes, P. J., Edwards, R., Damore, D., & Scott, D. (2009). Glaciers as a source of ancient and labile organic matter to the marine environment. *Nature*, 462(7276), 1044–1047. https://doi.org/10.1038/nature08580
- McClelland, J. W., Holmes, R. M., Peterson, B. J., Raymond, P. A., Striegl, R. G., Zhulidov, A. V., et al. (2016). Particulate organic carbon and nitrogen export from major Arctic rivers. *Global Biogeochemical Cycles*, *30*, 629–643. https://doi.org/10.1002/2015GB005323
- Olefeldt, D., Persson, A., & Turetsky, M. R. (2014). Influence of the permafrost boundary on dissolved organic matter characteristics in rivers within the Boreal and Taiga plains of western Canada. *Environmental Research Letters*, 9(3). https://doi.org/10.1088/1748-9326/9/ 3/035005
- Quinton, W. L., Hayashi, M., & Chasmer, L. E. (2011). Permafrost-thaw-induced land-cover change in the Canadian subarctic: Implications for water resources. *Hydrological Processes*, 25(1), 152–158. https://doi.org/10.1002/hyp.7894
- Regnier, P., Friedlingstein, P., Ciais, P., Mackenzie, F. T., Gruber, N., Janssens, I. A., et al. (2013). Anthropogenic perturbation of the carbon fluxes from land to ocean. *Nature Geoscience*, 6(8), 597–607. https://doi.org/10.1038/ngeo1830
- Singer, G. A., Fasching, C., Wilhelm, L., Niggemann, J., Steier, P., Dittmar, T., & Battin, T. J. (2012). Biogeochemically diverse organic matter in Alpine glaciers and its downstream fate. *Nature Geoscience*, 5(10), 710–714. https://doi.org/10.1038/ngeo1581
- Snowdon, L. R., Brooks, P. W., Williams, G. K., & Goodarzi, F. (1987). Correlation of the Canol Formation source rock with oil from Norman Wells. Organic Geochemistry, 11(6), 529–548. https://doi.org/10.1016/0146-6380(87)90008-8
- Stibal, M., Šabacká, M., & Žárský, J. (2012). Biological processes on glacier and ice sheet surfaces. Nature Geoscience, 5(11), 771–774. https:// doi.org/10.1038/ngeo1611
- Turetsky, M. R., Amiro, B. D., Bosch, E., & Bhatti, J. S. (2004). Historical burn area in western Canadian peatlands and its relationship to fire weather indices. *Global Biogeochemical Cycles*, 18, GB4014. https://doi.org/10.1029/2004GB002222
- Vonk, J. E., Drenzek, N. J., Hughen, K. A., Stanley, R. H. R., McIntyre, C., Montluçon, D. B., et al. (2019). Temporal deconvolution of vascular plant-derived fatty acids exported from terrestrial watersheds. *Geochimica et Cosmochimica Acta*, 244, 502–521. https://doi.org/ 10.1016/j.gca.2018.09.034
- Wacker, L., Bonani, G., Friedrich, M., Hajdas, I., Kromer, B., Němec, M., et al. (2010). Micadas: Routine and high-precision radiocarbon dating. Radiocarbon, 52(2), 252–262. https://doi.org/10.2458/azu_js_rc.52.3660
- Welte, C., Hendriks, L., Wacker, L., Haghipour, N., Eglinton, T. I., Günther, D., & Synal, H. A. (2018). Towards the limits: Analysis of microscale ¹⁴C samples using EA-AMS. *Nuclear Instruments and Methods in Physics Research, Section B: Beam Interactions with Materials and Atoms*, 437(January), 66–74. https://doi.org/10.1016/j.nimb.2018.09.046